Author's response to reviews

Title: The Alberta Ambassador Program: delivering Health Technology Assessment results to rural practitioners

Authors:

Saifudin Rashiq (srashiq@ualberta.ca)
Pamela Barton (pamela.barton@calgaryhealthregion.ca)
Christa Harstall (christa.harstall@ahfmr.ab.ca)
Donald Schopflocher (donald.schopflocher@gov.ab.ca)
Paul Taenzer (paul.taenzer@calgaryhealthregion.ca)

Version: 5 Date: 24 March 2006

Author's response to reviews:

A Conclusions section has been inserted after the Discussion.

Authors' contributions have been deleted from the title page, and placed after the conclusions section, in a section titled Authors' contributions.

Acknowledgements - The text "This project was sponsored by a Canadian Coordinating Office on Health Technology Assessment Capacity Building Grant to Dr. Paul Taenzer." has been removed from the title page.